

Received 7 June 2024, accepted 1 September 2024, date of publication 10 September 2024, date of current version 6 November 2024. Digital Object Identifier 10.1109/ACCESS.2024.3457238

RESEARCH ARTICLE

Measuring and Promoting Self-Regulated Learning Using Spaced Questionnaires

AMR ADEL ELSAYED^(D)^{1,2}, MANUEL CAEIRO-RODRÍGUEZ^(D)², (Senior Member, IEEE), FERNANDO A. MIKIC-FONTE^(D)², (Senior Member, IEEE), ADRIÁN LUGILDE-LÓPEZ^(D)², (Member, IEEE),

AND MARTÍN LLAMAS-NISTAL^{®2}, (Senior Member, IEEE)

¹Arab Academy for Science, Technology & Maritime Transport, Cairo 21532, Egypt

²atlanTTic Research Center for Telecommunication Technologies, Universidade de Vigo, 36310 Vigo, Spain

Corresponding author: Amr Adel ElSayed (aadel@uvigo.es)

This work has received financial support from Consellería de Educación, Ciencia, Universidades e Formación Profesional and co-funded by UE, and from the Galician Regional Government under project GPC-ED431B 2023/37.

This work involved human subjects or animals in its research. Approval of all ethical and experimental procedures and protocols was granted by the Arab Academy for Science, Technology and Maritime Transport.

ABSTRACT Self-Regulated Learning addresses key students' competences for performance in higher education, particularly in engineering disciplines. Assessing these competences helps identify students' gaps and provide appropriate support towards their development. Currently, the main self-regulated learning assessment tools are questionnaires with a large number of items administered in one session, for example, the MSLQ has 81 questionnaire items. Rather than snapshotting the students' self-regulated learning competences in a single session, this manuscript proposes a novel assessment approach, the SMSRLQ, which includes 150 questionnaire items (81 from the MSLQ and 69 from other questionnaires) but spacing the items weekly and providing feedback monthly. The spaced approach was experienced on several Egyptian universities using a social network service (Facebook) and distributing students into control (113) and experimental groups (107). Many variations in the students' answers are observed, with statistically significant differences between 11% to 30% in MAE and MAPE indicators. Variations in standard deviation, reliability and clustering between the approaches also were identified, demonstrating that there are clear differences between the two administration approaches. Regarding monthly feedback, the results show that it enhances students' participation and improves their self-regulated learning competences: 77% of the experimental group respondents found that their self-learning knowledge was developed after the study, compared to 50% of the control group respondents.

INDEX TERMS Higher education, self-regulated learning, spaced questionnaires.

I. INTRODUCTION

Self-Regulated Learning (SRL) is a "self-directive process by which learners transform their mental abilities into academic skills" [1]. It is a comprehensive term that encompasses multiple components, like self-efficacy, volition, cognition, metacognition, and motivation [2]. SRL is a transversal topic that can be applied to various academic

The associate editor coordinating the review of this manuscript and approving it for publication was Antonio J. R. Neves¹⁰.

disciplines including the field of engineering education. According to [3], self-regulated learners are the "*masters of their personal learning activities*". They are active, self-aware, knowledgeable, and capable of planning, setting goals, organizing, scheduling, self-monitoring, self-evaluating, and deciding their proper learning approach. Learners adopt different SRL strategies according to contextual factors like self-belief, social support, cultural background, and the subject natures. However, the basic elements of SRL are the same [4], [5], [6], [7], [8].

Equipping higher education students with a balanced combination of SRL competences is an essential element for them to achieve their learning objectives, evaluate their progress, and finish their college education successfully [9], [10], [11]. The difference in students' SRL competences can explain their different learning progress and academic accomplishment [12], [13]. Nevertheless, the appropriate development of SRL competences can be problematic because it involves many components: motivation, emotion regulation, goal management, time management, task value and understanding, environment structuring, resource management, help seeking, peer learning, evaluating, monitoring, learning strategies, etc. This can be particularly challenging for first year University students, as they face difficulties in gaining experience and being acquainted with the University life during the transitional period between high school to higher education [14], [15]. Moreover, the students' need to enhance their SRL capacities has become more urgent after the outbreak of the COVID-19 pandemic [16], [17]. The pandemic outbreak resulted in an increasing adoption of online learning. This mode of learning demands higher SRL capacities to achieve the learning outcomes.

The lack of students' self-regulation can cause suboptimal performance and the use of SRL strategies can predict dropout rate [18], [19], [20]. This problem is obvious in engineering education that witnesses a high dropout rate of students [21], [22]. For example, the dropout rate among the Spanish students in the discipline of Engineering and Architecture reached a significantly high level of 36%. Most of the dropouts were among the novice first year students [21]. In addition, SRL can also contribute to support the development of students' soft skills, which are highly required for the engineering graduates entering the industry.

The good news is that SRL promotion can occur gradually via learning and practice [23], [24]. Different frameworks [16], [25], [26], exist to analyze, recognize, and support learners' SRL development. Among all the elements involved in SRL competences development, assessment of the actual student competences is a key one [27]. Nowadays, self-report instruments are the most popular ones. They are provided either as large questionnaires including questions for the bulk of the SRL components, or as shortened questionnaires considering just some components. In both cases, these questionnaires are usually administered in a unique session to measure students SRL skills.

Currently, the most popular SRL questionnaire is the Motivated Strategies for Learning Questionnaire (MSLQ). This includes 81 questions, but it is also available in shortened versions with less than half of these questions. Despite psychometric problems [28], MSLQ is a benchmark in measuring the application of students' strategies to be self-regulated [29]. This approach has shortcomings:

1) For large questionnaires, respondents may tend to answer carelessly or even quit answering.

- 2) For reduced questionnaires, they also reduce the coverage of the variety of SRL competences.
- One-time questionnaires provide a snapshot instead of a dynamic view about the learners' SRL development over time [30].
- 4) Questionnaires are susceptible to multiple biases that negatively affect the reliability of the gathered data [31]. For instance, students may overestimate their SRL capacities. Existing research has shown discrepancies between their self-reported data and their actual behaviors [32], [33].

Considering these previous shortcomings, we envisioned a novel approach to measure and promote SRL competences: spacing questionnaire items over time. This approach is based on dividing a large questionnaire into small chunks and administering questions on a spaced basis. As the questionnaire period is prolonged, it can include a larger number of questions. Particularly, we propose the Social Media Self-Regulated Learning Questionnaire (SMSRLQ), involving 150 questions, 81 ones from the original MSLQ [34]. In addition, students receive monthly feedback as an SRL competence report based on the collected answers.

For this new spaced approach, the following research questions are considered:

- RQ1: Do differences exist due to the two administration approaches? We wonder if the students' answers vary between the single session and the spaced administration and if there is a pattern in such variation.
- RQ2: Does the feedback provision enhance the students' participation and cause a difference in their SRL competences development? We wonder if the provision of feedback to students can enhance their participation and cause any difference in their SRL competences development.

The structure of the rest of the manuscript is as follows: Section II reviews the related literature. The third section discusses the used data gathering instruments, the study design, and the methodology. Section IV presents the results of the SMSRLQ and compares them against the MSLQ results, considering specially the questions in the SMSRLQ taken from the MSLQ. Section V discusses the results against the proposed research questions. Finally, section VI includes the conclusions, limitations, contributions, implications, and future work suggestions.

II. LITERATURE REVIEW

During the last decades, the focus on learning and pedagogical theories has shifted from external regulation by instructors to the empowerment of students as main characters. Particularly, a plethora of prior research stressed the constructive influence of SRL on the higher education students' academic achievement. It showed that self-regulated learners surpass their counterparts who lack SRL skills [35], [36]. SRL practices endorse adaptation to novel contexts, tasks, and situations, and therefore can lessen the effect of

TABLE 1. Previous work contributions.

Scientific contribution	Reference(s)
Demonstrating that self-regulated learners surpass their counterparts who lack SRL skills	[35], [36]
Demonstrating that SRL practices can lessen the effect of surprising circumstances such as COVID-19 on the students' academic	[37]
performance	
Demonstrating that more successful learners are engaged in greater and more complicated SRL behaviors than other students.	[38], [39]
Demonstrating that the improvement in the students' SRL partially mediates their achievement	[40]
Demonstrating that the difference between high achievers and low-achieving students lies principally in the quality rather than the	[41]
quantity of the utilized SRL strategies	
Demonstrating the shortcomings of self-report instruments in assessing the learners' SRL.	[42], [43]
Demonstrating the recent trend towards data analytics approaches due to the self-report instruments shortcomings.	[42], [44], [45]
Demonstrating that the instruments based on data analytics try to track and directly analyze the genuine experience of learners, rather	[46]
than the learners' perceptions	
Demonstrating that SRL measurement by data analytics can take place either solely, or with other self-report measures to obtain data	[42], [44], [47]
triangulation, recognize the different students' SRL behaviors more exactly, comprehensively understand the contextual effects,	
guide students, and better forecast their performance.	
Examining the evolution of engineering students' SRL and its positive influence on their achievement.	[48-50]
Demonstrating the limitations of data analytics techniques in SRL assessment.	[30], [51]
Demonstrating the advantages of Questionnaires in SRL assessments and their usage in various learning environments.	[30], [52]- [56]
Demonstrating the significance of properly designed SRL interventions.	[57-61]
Introducing the third wave of SRL assessment that combines measurement and intervention.	[51]
Guidelines for the provision of effective and efficient interventions in higher education settings.	[40]

surprising circumstances such as COVID-19 on the students' academic performance [37]. More successful learners are engaged in greater and more complicated SRL behaviors than other students [38], [39]. On the contrary, [40] found that the improvement in the students' SRL partially mediates their achievement and [41] found that the difference between high-achieving and low-achieving students lies principally in the quality rather than the quantity of the utilized SRL strategies.

A great proportion of previous research has focused on using self-report instruments for SRL assessment. Of course, these instruments rely on the self-assessment information provided by learners. Such information is susceptible to bias and can be inaccurate because of factors such as the learners' memory retrieval capacity, perceptions, and points of view. Moreover, these measurements do not present a dynamic view of the learners' SRL evolution through the learning process [42], [43].

Due to the self-report instruments shortcomings, there is a recent trend towards data analytics approaches [42], [44], [45]. These instruments try to track and directly analyze the genuine experience of learners, rather than the learners' perceptions [46].

SRL measurement by data analytics can take place either solely, or with other self-report measures to obtain data triangulation, recognize the different students' SRL behaviors more exactly, comprehensively understand the contextual effects, guide students, and better forecast their performance [42], [44], [47]. In the engineering context, [48] examined the evolution of students' SRL profiles over time using longitudinal clustering, the results of [49] revealed better performance and higher achievement of highly self-regulated engineering students in comparison to their minimal selfregulated colleagues, and [50] showed a positive relationship between an SRL based approach and engineering students' success. To the best of the authors' knowledge, the application of data analytics techniques has been solely in lab experiences. In addition, they have limitations in observing or monitoring the students' offline activities and behaviors [30], [51].

Therefore, despite criticisms, questionnaires are the most suitable option to assess SRL competences for higher education students. They are more feasible technically, economically, and easier to apply for large-scale assessments [30]. Questionnaires provide valuable insights about the students' SRL perceptions, beliefs, attitudes, and behaviors at a coarse-grained global level [52], [53]. We cannot neglect the learners' perceptions as they offer vital information for investigating SRL [54]. Information provided by questionnaires is exceedingly difficult or even impossible to obtain by other ways, particularly learning analytics [55]. In addition, despite questionnaires being principally for face-to-face classrooms, their usage involves also measuring SRL in online learning environments [56].

Regarding SRL intervention, a plethora of studies have emphasized the significance of intervention to improve the achievement of students; particularly those with low SRL levels [57], [58]. Several intervention methods have been tested to foster the students' SRL. The intervention can take place implicitly or explicitly. Despite achieving better academic achievements, the application of explicit SRL interventions is rare compared to implicit interventions [59], [60].

Many authors have demanded the need for a third wave of SRL assessment combining measurement and

Scale	Components	Category	Description
Motivation	Value components	Intrinsic goal-orientation	The degree of students' perception to their participation in academic tasks as an end all to itself, rather than a means to an end
		Extrinsic goal- orientation	The degree of students' perception to their participation in academic tasks as a means to an end
		Task value	The degree to which students perceive learning tasks as interesting, significant, and beneficial
Expectancy components		Control of learning beliefs	The degree to which students believe that their efforts yield positive results
		Self-efficacy for learning and performance	The degree of students' ability to motivate themselves and control their emotions
	Affective components	Test anxiety	The degree to which students possess negative emotions and thoughts about tests and exams
Learning	Cognitive and	Rehearsal	The degree of using strategies like rereading and memorizing
Strategies	metacognitive strategies	Elaboration	The degree of using strategies like summarizing and paraphrasing material
		Organization	The students' capacity to identify and organize the main course ideas
		Critical thinking	The degree to which students compare their performance against criteria like previous performance, others' performance, or an absolute performance level
		Meta-cognitive self- regulation	The degree to which students observe how they progress toward achieving their learning objectives [1]
	Resource management strategies	Time and study environment	The goodness of students' use of their time, schedule, and study place
		Effort-regulation	The degree of students' dedication to manage their efforts to accomplish their goals against various difficulties. The degree of using strategies like rereading and memorizing
		Peer learning	The degree to which students collaborate with their counterparts to achieve their goals
		Help-seeking	The degree to which students ask for assistance from peers and instructors to achieve their goals

TABLE 2. MSLQ scales, components, and categories based on [74].

intervention [42]. The intervention design must be proper to obtain a desirable effect, otherwise the students' dissatisfaction will ruin its effect [61]. The intervention design can follow guidelines like those of [40] for the provision of effective and efficient interventions in higher education settings. Table 1 presents a summary of the scientific contributions in the reviewed literature.

III. METHODS

A. INSTRUMENTS

The study involved two questionnaires:

- MSLQ is the most used questionnaire to measure SRL learners' competencies in higher education settings [62], [63]. It includes 81 items divided into two major scales, 5 components and 15 categories (see Table 2).
- SMSRLQ is composed of 150 items including the 81 MSLQ items and 69 items from 11 other questionnaires: Metacognitive Awareness Inventory (MAI) [64], Patterns of Adaptive Learning Scales (PALS) [65], Regulation of Learning Questionnaire (RLQ) [54], Self-Regulated Knowledge Scale-University (SRKS-U) [66], Self-Regulated Learning Questionnaire (SRLQ) [67], Self-Regulation Strategy Inventory–Self-Report Questionnaire (SRSI-SR) [68], Student Learning Strategies Questionnaire (SLSQ) [69], Students' Adaptive Learning Engagement in Science (SALES) [70], Survey of Academic Self-Regulation (SASR) [71], Survey of

Self-regulated Learning with Technology at the University (SRLTU) [45], and Revised Two Factor Study Process Questionnaire (R-SPQ-2F) [72] (see Table 3). REVERSED questions from the MSLQ were taken as indicated, including the Test Anxiety category questions.

The SMSRLQ items were spaced so that only 3 questions were presented to each participant daily. The spaced repetition learning technique was selected because it includes repetitive long inter-trial intervals, and therefore leads to more robust memory [73]. SMSRLQ contents are displayed in Appendix A (see Table 9). In this manuscript we use mqSMSRLQ to refer to the answers collected on the SMSRLQ corresponding to the 81 MSLQ questions.

Both questionnaires were in two languages, English and Arabic, to encourage the participation of none-English speakers. The questionnaires' responses were based on a five-point Likert scale ranging from "1" which corresponds to "Strongly Disagree" to "5" which corresponds to "Strongly Agree".

B. CONTEXT, PARTICIPANTS, AND PROCEDURE

Students participating in this experience belong to three colleges in two different Egyptian Universities. The majority (416) belong to the College of Management & Technology within the Arab Academy for Science, Technology & Maritime Transport (AASTMT). The minority are students in the

 TABLE 3. Number of SMSRLQ items taken from other questionnaires.

Questionnaire	Intrinsic goal-orientation	Extrinsic goal-orientation	Task value	Control of learning beliefs	Self-efficacy for learning and performance	Test anxiety	Rehearsal	Elaboration	Organization	Critical thinking	Meta-cognitive self-regulation	Time and study environment	Effort-regulation	Help-seeking	Peer-learning	Total number of items
MSLQ	4	4	6	4	8	5	4	6	4	5	12	8	4	4	3	81
MAI	1			1			1	1	1	1	10	1				17
PALS	2	2		1		1										6
RLQ			1										1			2
SRKS-U							1									1
SRLQ											2					2
SRSI-SR					1							4		2		7
SLSQ	1	1		1	1						2		1			7
SALES			3							1		1				5
SASR			1	1	2	2			1		1	1		1		10
SRLTU								1	2						4	7
R-SPQ-2F		1		1	1					1		1				5
Total	8	8	11	9	13	8	6	8	8	8	27	16	6	7	7	150

College of Languages & Communication in the AASTMT (11) and the Faculty of Nursing in Helwan University (51). All participants were seeking bachelor's degrees in the previously mentioned colleges and universities.

The study procedure is displayed in Fig. 1. The MSLQ administration took place in a single session at the beginning of the academic semester 2021/2022, during the inaugural lectures. Students were also introduced into the whole study procedure. The students participating in this initial session (N=478) read and signed an informed consent. 383 participants filled in physical pen and paper questionnaires, and the rest filled it electronically. This initial stage was conducted between October 23, 2021, and November 5, 2021. After filling in the questionnaire, the participants were asked to set up their Facebook accounts for the second stage.



FIGURE 1. Study procedure.

The SMSRLQ was applied to the students who voluntarily continued participation online (N=220). In this way, MSLQ and the SMSRLQ were compared to answer RQ1.

Regarding RQ2, we randomly assigned the students involved in the SMSRLQ in two groups. An experimental group named "SRL 2021-2022 (1)" (N=107) was provided with feedback every month and comprehensive feedback about their SRL performance at the end of the study period. A control group named "SRL 2021-2022 (2)" (N=113) did not receive any feedback.

The SMSRLQ stage extended over a 10-week period via Facebook from November 6, 2021, to January 7, 2022, excluding the mid-term exam period between November 20 and November 26, 2021. The SMSRLQ questions distribution was every two days in the early morning. Triggering mobile notifications in the early morning rather than the evening or unexpected moments leads to better SRL results as indicated by [75]. Every question was in the form of a Facebook poll (see Fig. 2). Facebook responses were collected manually and then entered into Excel files for further processing.

As participation decreased with time, we eliminated the responses of 136 participants: 61 from the experimental group and 75 from the control group. As a result, there was an uneven distribution of students, according to gender and academic year, among the experimental and control groups. 70% of the experimental group participants were female students in comparison to a mere 26% of the control group participants. 63% of the experimental group participants were novice students in comparison to 45% of the control group participants. Table 4 summarizes the demographic data of participants in both approaches.

ا ask myself if I have considered all options when solving a problem أسأل نفسي إذا كنت قد فكرت في المعاد ما مسألة ما جميع الخيارات عند حل مسألة ما



FIGURE 2. SMSRLQ question poll in english and arabic.

TABLE 4. Participants' demographics.

	MSLQ	SMSRLQ	SMSRLQ	SMSRLQ
	N=478	N=84	(Experimental	(Control
			group): N=46	group):
				N=38
Gender	Males:	Males:	Males: 30%	Males:
	61%	50%	Females: 70%	74%
	Females:	Females:		Females:
	39%	50%		26%
Academic	1 st and 2 nd			
year	year:	year: 55%	year: 63%	year: 45%
	56%	Elder: 45%	Elder: 37%	Elder: 55%
	Elder:			
	44%			
Age	$19.46 \pm$	19.33±1.44	19.07±1.36	19.66±1.49
	1.63			
Number	6.04 ± 0.65	6.17±0.51	6.11±0.43	6.24±0.59
of subjects				
in the term				

After the experience, we excluded the data of those students who did not participate. To do this, students with more than 50 missing questions were removed. The closing number of students was 84: 46 students in the experimental group and 38 students in the control group. In addition, questions over 130 were removed because their percentage of missing answers was over 10% [76]. This can have a significant effect on the "Test anxiety" and "Effort regulation" categories as the number of MSLQ questions reduced from 4 to 2, namely 50%. For the rest of the students and missing answers it was performed an imputation process, using the Expectation-Maximization (EM) algorithm, provided by the IBM SPSS Version 29.0.1.0 (171). The total number of imputations was 610 for the 130 questions and 84 students (6%).

Depending on the mean score of the experimental group participants, the feedback was categorized into low, medium, and high categories for each SMSRLQ competence. Three feedback events were provided: the first by the end of the first month of the study, the second by the end of the second month, and the final feedback after the end of the study. The feedback, whose content is adapted from [74] and [77], was in the form of a personalized Facebook message for each active participant in the experimental group. While the 1st and 2^{nd} monthly feedback events evaluate the students' responses and provides recommendations only for the questions asked during the 1^{st} and 2^{nd} months of the study respectively, the final feedback evaluates the students' overall performance and offers final recommendations. The feedback includes 3 sections:

1. Section I includes a summary of the feedback's goal and structure: how to interpret the scores, the participant's mean score in relation to the mean score of all participants, the participant's number of responses, and response rate in relation to the average response rate of all participants. Moreover, the final feedback compares the participant's response rate in the first and second months of the study.

2. Section II includes a thorough analysis of the participant's performance in every SRL category, and suggestions based on the mean score in each category.

3. Section III concludes the participant's performance and provides final recommendations.

IV. RESULTS

A. DESCRIPTIVE STATISTICS

Initially, items marked as "reversed" were reversed subtracting the original score from 6. Then, students' answers from the initial MSLQ and the spaced SMSRLQ were processed to calculate the mean for each student per SRL category, (see Table 5), considering the values of the answers given by each student. This mean value is considered an indicator of the level of SRL competency of the student in such category. The mqSMSRLQ represents the results of the MSLQ answers collected on a spaced basis during the administration of the SMSRLQ.

The columns 2 to 4 of Table 5 show the mean and standard deviation of all the students SRL competencies in such category. Surprisingly, the standard deviation values are much lower in the SMSRLQ for all SRL competences. Indeed, standard deviation is double for the MSLQ than for the SMSRLQ. In any case, the most significant differences are in comparing the mqSMSRLQ to the MSLQ, as they are the same questions answered by the same students and the standard deviation is lower for the mqSMSRLQ in all the cases. This could be a consequence of the imputation performed on the SMSRLQ missing answers. Maximum and minimum mean values in each of these two columns are in bold font. The mqSMSRLQ has 8 maximum values and MSLQ has 6. Self-efficacy has the same result.

The last four columns of Table 5 show the Mean Average Error (MAE) and Mean Average Percentage Error (MAPE) comparing the SMSRLQ vs. the MSLQ and the mqSMSRLQ vs. the MSLQ. All MAE differences are greater than 0.4 and 11%. This shows major differences in the students' answers depending on how they are collected. A difference in the first comparison can be accepted, as the questions in both instruments are different. Nevertheless, there is also a similar difference in the second comparison, and this is a bit weird, as the questions are exactly the same. The difference

SRL category	SMSRLQ	mqSMSRLQ	MSLQ	MAE (SMSRLQ- MSLQ)	MAPE (SMSRLQ- MSLO) %	MAE (mqSMSRLQ- MSLO)	MAPE (mqSMSRLQ- MSLO) %
Intrinsic goal- orientation	4.22±0.31	3.95±0.37	4.03 ±0.62	0.46	11.08	0.47	12.09
Extrinsic goal- orientation	3.75±0.47	4.17±0.48	4.18 ±0.59	0.62	17.42	0.46	11.24
Task value	3.84 ± 0.38	4.14 ± 0.42	4.12 ± 0.58	0.56	14.84	0.50	12.27
Control of learning beliefs	3.70± 0.28	3.67± 0.40	3.71 ± 0.58	0.47	12.67	0.47	13.05
Self-efficacy	3.93 ± 0.36	3.99± 0.34	3.99 ± 0.60	0.49	12.67	0.47	12.02
Test anxiety	3.24 ± 0.80	3.52 ± 0.87	2.88 ± 0.97	0.75	24.16	0.90	26.60
Rehearsal	4.05± 0.43	4.10 ± 0.42	4.09± 0.64	0.46	12.21	0.47	12.32
Elaboration	4.11± 0.35	4.08 ± 0.41	4.04 ± 0.41	0.51	12.76	0.53	13.53
Organization	3.66 ± 0.48	3.65 ± 0.56	3.94 ± 0.77	0.65	18.58	0.68	20.19
Critical thinking	3.86± 0.34	3.92 ± 0.35	3.87± 0.68	0.51	13.37	0.49	12.83
Metacognitive self- regulation	3.91± 0.34	3.88 ± 0.36	3.76± 0.58	0.46	11.80	0.42	11.06
Time and study environment	3.27± 0.37	3.38± 0.47	3.64 ± 0.60	0.52	16.38	0.47	14.75
Effort regulation	3.67 ± 0.45	3.88 ± 0.46	3.48± 0.73	0.59	16.37	0.64	16.59
Help-seeking	4.03 ± 0.42	4.03 ± 0.54	3.60± 0.67	0.63	15.61	0.66	16.28
Peer-learning	2.78 ± 0.52	3.07± 0.54	3.35 ± 0.78	0.82	31.09	0.61	20.75

is quite large in some categories (in bold), over 20% for: Test anxiety, Organization and Peer-learning. Differences in Test anxiety could be a consequence of the reduction of questions from 4 to 2. The same statistics were calculated considering the control/experimental group, gender, and academic year, but they provided analogous results. In any case, when comparing the results according to gender and academic year across the experimental and control groups, we must consider the uneven distribution of participants. As explained previously, this was not due to a careless study design, but a result of the uneven participation and involvement of students in the experimentation.

B. INFERENTIAL STATISTICS

Considering the previous results, we conducted ANOVA [78] and t-student [79] to statistically evaluate the significant differences in the SRL categories. Table 6 shows the p-value results of the analysis involving the SMSRLQ, mqSMSRLQ and MSLQ. Numbers in bold indicate the existence of significant differences in several categories as they are lower than 0.05.

As previously mentioned, we can easily explain the differences between SMSRLQ and MSLQ as they include different questions. Nevertheless, we also observe relevant differences between mqSMSRLQ and MSLQ (last column). Particularly, the last 5 rows including the "Resource Management Strategies" component and the "Metacognitive self-regulation" category show significant differences. Again, the "Test Anxiety" also shows a significant difference, but this could be a consequence of the reduction of questions from 4 to 2.

C. RELIABILITY

Cronbach's alpha coefficient [80] was used to evaluate the reliability and internal consistency of the instruments, considering also the MSLQ questions performed in the SMSRLQ spaced questionnaire (see Table 7). The values under 0.500 are in bold, indicating they are under the acceptable threshold for reliability. As observed, all the three instruments have values under this threshold: 6 for the SMSRLQ, 8 for the mqSMSRLQ and 2 for the MSLQ. Again, it is quite surprising that the mqSMSRLQ and the MSLQ provide so different results.

However, higher Cronbach's values are not a guarantee of better reliability. Reliability measures the test scores not the test itself and is dependent upon the tested sample [81], [82]. In this way, the lower values on the mqSMSRLQ may be because of spaced questions. This means that the reliability of students' answers decreases when they answer over time.

The MSLQ values are higher in all categories. Nevertheless, for the "Test anxiety," "Time and study environment" and "Peer learning" the SMSRLQ provides better results.

TABLE 6.	ANOVA and	t-student	results	p-values.
----------	-----------	-----------	---------	-----------

SRL category	ANOVA (SMSRLQ, mqSMSRL Q, MSLQ)	t-student (SMSRLQ, mqSMSRL Q)	t-student (SMSRL Q, MSLQ)	t-student (mqSMSRL Q, MSLQ)
Intrinsic goal- orientation	0.020	E-18	0.003	0.237
Extrinsic goal- orientation	E-12	E-24	E-09	0.833
Task value	E-05	E-26	E-05	0.708
Control of learning beliefs	0.805	0.226	0.952	0.592
Self- efficacy	0.609	0.002	0.353	0.910
Test Anxiety	0.001	E-04	E-03	E-05
Rehearsal	0.827	0.013	0.565	0.924
Elaboration	0.692	0.029	0.371	0.667
Organizatio n	0.002	0.601	0.001	0.001
Critical thinking	0.704	0.581	0.408	0.477
Metacogniti ve self- regulation	0.063	0.007	0.013	0.041
Time and study environmen t	E-06	E-05	E-08	E-05
Effort regulation	E-05	E-06	0.014	E-06
Help- seeking	E-07	0.841	E-08	E-07
Peer- learning	E-08	E-11	E-09	E-04

D. CLUSTERING

Clustering is an unsupervised data mining technique used to describe data into meaningful or useful groups [83]. The K-means algorithm [84] was applied to cluster the differences in the mean values in each category among all instruments. Fig. 3 shows a box and whisker plot for the differences between the mqSMSRLQ and the MSLQ. In this figure, there is a displacement in almost all categories towards lower values. There are two categories doing the opposite: "Text anxiety" and "Effort regulation."

Taking the differences in all categories, two clusters were found by the elbow method. One cluster related to the students who progressed in almost all the categories. The other related to those with deteriorated values. 10 out of 12 (83%) students of the "progressed" cluster belong to the experimental group.

We repeated the cluster analysis excluding the "Test anxiety" and "Effort regulation" categories, as they can introduce some kind of bias because their questions were reduced from 4 to 2. In this case, the number of clusters provided by the elbow method was 3. Fig. 4 displays the cluster centers. Cluster 2 shows progress in all categories, while

TABLE 7.	Cronbach	's alpha	values	of the	three	variants.
----------	----------	----------	--------	--------	-------	-----------

SRL Category	SMSRLQ	mqSMSRLQ	MSLQ
Intrinsic goal orientation	0.401	0.259	0.587
Extrinsic goal orientation	0.435	0.182	0.341
Task value	0.763	0.622	0.770
Control of learning beliefs	0.154	0.129	0.395
Self-efficacy	0.707	0.540	0.862
Test anxiety	0.836	0.734	0.799
Rehearsal	0.510	0.017	0.647
Elaboration	0.550	0.550	0.790
Organization	0.596	0.611	0.722
Critical thinking	0.468	0.129	0.760
Meta-cognitive self-regulation	0.783	0.616	0.822
Time and study environment	0.696	0.532	0.637
Effort regulation	0.307	0.483	0.643
Help-seeking	0.390	0.468	0.571
Peer-learning	0.619	0.294	0.559

cluster 1 shows deterioration in "Organization" and "Time and student environment", and cluster 3 deteriorates in "Peer learning" and "Help seeking". The ANOVA analysis of this clustering supplies values under 0.01 for all the categories, meaning they are all significant.

Table 8 shows the students assigned to each of the three clusters. Most students of cluster 2 belong to the experimental group. Meanwhile, control group students belong to cluster 1, and experimental group students are assigned also to cluster 3.

E. SATISFACTION QUESTIONNAIRE

Following the data collection, a satisfaction questionnaire adapted from [85] was administered to both groups to assess the participants' general satisfaction with the study concerning the interaction, the instruction, the instructor, and the used technology. 23 students participated voluntarily and anonymously: 13 from the experimental group and 10 from the control group. The questionnaire included 16 items: 13 items validated against a 5-Likert scale and 3 exploratory questions. The satisfaction questionnaire contents are displayed in Appendix B (see Table 10).

The responses to the Likert scale items are positive in general. Nonetheless, the acceptance levels of the experimental group respondents were higher in nearly all aspects than those of the control group respondents (see Fig. 5).

77% of the experimental group respondents, in comparison to 60% of the control group respondents, stated that the study kept them alert and focused. Keeping students thinking about their SRL behaviors can be related to the





FIGURE 3. Box and whisker plot for the differences between means of mqSMSRLQ and MSLQ.



FIGURE 4. Cluster centers for the differences between the mqSMSRLQ and MSLQ excluding "Test anxiety" and "Effort regulation."

spaced repetition learning technique that includes repetitive long inter-trial intervals, and therefore leads to more robust memory [73]. 85% of the experimental group respondents were satisfied with the interaction with the lecturer and found that the feedback was timely throughout the study period, in comparison to 70% of the control group respondents. Only 70% of the control group respondents were pleased with their participation in the study, while all the experimental group respondents were satisfied. 85%-90% of respondents from the two groups think the study encouraged them to assess the way they learn and study. 77% of the experimental group respondents found that their self-learning knowledge developed after the study, compared to 50% of the control group respondents. This variation can be due to the influence of interaction with the lecturer and feedback. 60% of the control group respondents (in comparison to 92% of the experimental group respondents) find the study instructions clearly communicated.

Regarding the exploratory questions (see Fig. 6), approximately 96% of the respondents in the two groups use social



FIGURE 5. Satisfaction questionnaire Likert scale responses across the experimental and control groups.



FIGURE 6. Satisfaction questionnaire exploratory questions responses in the experimental and control groups.

 TABLE 8. Number of students of the experimental and control groups in the 3 clusters.

Group	Cluster 1 (N=34)	Cluster 2 (N=25)	Cluster 3 (N=25)
Experimental	7 (21%)	19 (76%)	20 (80%)
Control	27 (79%)	6 (24%)	5 (20%)

media platforms daily. Facebook is the main social media platform for 60% of the control group respondents and 54% of the experimental group respondents. 100% of the experimental group respondents favor questionnaires administration via Facebook instead of the traditional pen and paper

questionnaires. This preference is 20% less in the control group respondents. This result is in line with responses to a Likert scale item; where 92% of the experimental group respondents view Facebook compatible for this study compared to 70% of the control group respondents. 85% of the experimental group respondents (compared to 70% of the control group respondents) have a tendency towards administering questionnaires on a spaced basis rather than a solitary session. On the one hand, some proponents of administering questionnaires on a spaced basis stated that "*it helped me already to know my strengths and weaknesses and highlight it while I'm studying already*", "*it will help me to recover*

information", "it keeps me engaged", "I am evolving and can change my perspective about certain things". On the other hand, some antagonists stated that "single session saves time and effort", "single session approach is more motivating, facilitate, and clarify questions", "my answer would be more honest", and "single session because sometimes I lost interest in the middle of the study, but your feedback motivated me to get back and answer the questions so you can evaluate me".

V. RESULTS DISCUSSION

The main motivation of this study is introducing a novel approach to measure and promote SRL adoption among higher education students via spacing questionnaire items and providing feedback over an extended period. The obtained results are compared against those of the popular MSLQ. In this section, we discuss the results against the proposed research questions.

RQ1: Do differences exist due to the two administration approaches?

The literature has identified a reciprocal relationship between the number of questions and the students' participation rate [34]. In this way, the distribution of questions in small chunks should improve participation. Nevertheless, the data provided does not support this statement, as the participation of students in the MSLQ was higher than in the spaced questionnaire. Of course, this result can be obvious and understandable, as the MSLQ answers were mainly collected in the context of a face-to-face section, while the SMSRLQ was administered online on a voluntary basis. In any case, the open answers to the satisfaction questionnaire also show how some students prefer to fill in the large questionnaire in a single session that spaced over time. Of course, answering questions over a certain period requires more effort and constancy, but learning also requires repetitive practice. Therefore, students may prefer a single session effort, but if learning comes from repetition, a spaced approach is more valuable.

Regarding RQ1, it is obvious that the performance of questionnaire items on a spaced basis introduces variations in responses between the MSLQ from one side, and the SMSRLQ and mqSMSRLQ from the other side. The variations between MSLQ and mqSMSRLQ are more relevant as both questionnaires involved exactly the same questions. These variations are clear in the high MAE and MAPE differences between the corresponding categories in the conventional MSLQ and mqSMRLQ. The lower standard deviation values of the SMSRLQ and mqSMSRLQ could indicate better reliability than the MSLQ [86]. The lower Cronbach's values of the SMSRLQ and mqSMSRLQ may be because of the questions spacing. Moreover, the p-value results of the ANOVA and t-student inferential statistics reveal the existence of variations between students' responses in the MSLQ and their responses in the mqSMSRLQ. Nevertheless, we cannot discard that these response variations between the instruments may be a result of a change in the SRL adoption by students over time. Despite this, their answers could also be more honest in one case than another. Recent research has shown that different ordering of questions can cause considerable changes in the respondents' attitudes [87]. The results provided in this manuscript confirm it considering not just changes in the ordering of questions, but also on their spacing along time.

RQ2: Does the feedback provision enhance the students' participation and cause a difference in their SRL competences development?

Regarding RQ2, the results show that the experimental group members participated more actively than their control group counterparts. Nevertheless, in both cases a decrease in participation with time was experienced. 46 experimental group members actively participated throughout the study period in comparison to 38 members of the control group. In general, students tend to modify their SRL behaviors over time [48], [88]. By tracking the cluster membership, we can explore the SRL evolution of students over time. 76% of the improving students in all SRL categories belong to the experimental group. Regarding the other two clusters, 79% of the students who deteriorated in "Organization" and "Time and student environment" belong to the control group. Oppositely, 80% of the students who deteriorated in "Peer learning" and "Help seeking" belong to the experimental group. Overall, 41% of the experimental group members belong to the "Progressed" cluster in comparison to a mere 16% of the control group members. The lower percentage of progressing participants in the control group could be related to the lack of feedback. These findings demonstrate the feedback influence on the SRL progression of the experimental group members.

When analyzing the satisfaction questionnaire results, the findings verify that the feedback provision results in a better SRL performance. The performance improvement via feedback is matching with a surplus of prior research such as [89] and [90]. Personalized feedback motivates students and keeps their concentration on their tasks [91]. Social comparison feedback benefits students, as they know how others perform [92].

VI. CONCLUSION

This manuscript introduces a spaced approach to measure and promote SRL among higher education students, including engineering students. It is intended to provide a comprehensive picture of the students' SRL development over time rather than a static snapshot at the semester beginning. This proposal can be considered in the context of the third wave of SRL assessment [42], involving both measurement and promotion. From a methodology point of view to the best of our knowledge, there is no other attempt to use spaced questionnaires for SRL promotion over an extended period.

The spaced approach can be very helpful to provide a more comprehensive view of SRL competences. As described in Section II and Table 2, SRL involves a variety of components

TABLE 9. SMSRLQ items.

1. I am confident I can manage my own learning (SASR)
2. I prefer course material that really challenges me so I can learn new
3. I come to most classes with questions in mind that I want answering
(R-SPQ-2F)
4. I have high academic standards for myself (SASR)
5. I want to do well in the courses I study because it is important to
show my ability to my family, friends, or others (MSLQ-30)
6. I know help is available if I need it (SASR)
(MSLO-7)
8. When I study, I set goals for myself in order to direct my activities in
each study period (MSLQ-78)
9. It is important for me to learn the course material (MSLQ-10)
10. I can motivate myself to learn when I need to (MAI)
11. I know how I will be graded (RLQ)
12. I have a regular place set aside for studying (MSLQ-65)
13. When I study, I practice saying the material to myself over and over (MSLO-39)
14. I stop and go back over new information that is not clear (MAI)
15. I pace myself while learning in order to have enough time (MAI)
16. Understanding the subject matter of the courses is very important to
me (MSLQ-27)
17. If I get confused taking notes in class, I make sure I sort it out afterwards (MSL 0-79)
18. If I try hard enough, then I will understand the course material
(MSLQ-18)
19. I summarize what I've learned after I finish (MAI)
20. It's important to me to improve my skills this year (PALS)
21. I try to identify students in the class whom I can ask for help if
necessary (MSLQ-75)
22. I'm certain I can master the skills being taught in different classes (MSI O-29)
23. I use the available study aids like chapter outlines (SASR)
24. When reading for courses, I make up questions to help focus my
reading (MSLQ-36)
25. The most satisfying thing for me is trying to understand the content
as morougnly as possible (MSLQ-22) 26. I try to work with other students from the class to complete the
course assignments (MSLQ-45)
27. When I study, I write brief summaries of the main ideas from the
readings and my class notes (MSLQ-67)
28. When studying, I try to determine which concepts I don't understand well (MSL Q-76)
29. Even if the work is hard, I can learn it (PALS)
30. When reading for the courses, I try to relate the material to what I
already know (MSLQ-64)
31. I memorize key words to remind me of important concepts in the
32. If course readings are difficult to understand I change the way I
read the material (MSLQ-44)
33. I ask myself if I have considered all options after I solve a problem
(MAI) 34. I make sure no one disturbs me when I study (SRSLSR)
25 When a theorem intermetation are a lain in the lain intermetation and th
35. When a theory, interpretation, or conclusion is presented in class or in the readings. I try to decide if there is good supporting suidence
(MSLO-47)
36. I find that at times studying gives me a feeling of deep personal
satisfaction (R-SPQ-2F)
37. I try to change the way I study in order to fit the course
requirements and the instructor's teaching style (MSLQ-56)
Vimeo (SRLTI)

TABLE 9. (Continued.) SMSRLQ items.

39. I'm certain I can understand the most difficult material presented in the readings for the courses (MSLO-6)
40. I expect to do well this semester (MSLQ-21)
41. It is the first time I have done this type of tasks (SLSQ)
42. I try to relate ideas in different courses whenever possible (MSLQ- 62)
43. Considering the difficulty of the courses, the teachers, and my skills, I think I will do well (MSLQ-31)
44. I spend a lot of my free time finding out more about interesting topics which have been discussed in different classes (R-SPQ-2F)
work (MSLQ-35)
46. I repeat several times the important things to be learnt (SRKS-U)
47. If I study in appropriate ways, then I will be able to learn the course materials (MSLQ-2)
48. Even when course materials are dull and uninteresting, I manage to keep working until I finish (MSLQ-74)
49. I make a schedule to help me organize my study time (SRSI-SR)
50. I'm confident I can learn the basic concepts taught in the courses (MSLQ-12)
51. Before I study new course material thoroughly, I often skim it to see how it is organized (MSLQ-54)
52. When I can't understand the course material, I ask another student in the class for help (MSLQ-68)
53. I try to understand the course material by making connections between the readings and the concepts from the lectures (MSLQ-69)
54. I make good use of my study time (MSLQ-43)
55. It is my own fault if I don't learn the course material (MSLQ-9)
56. When I study, I go through the readings and my class notes and try to find the most important ideas (MSLQ-42)
57. When I become confused about something I'm reading for class, I go back and try to figure it out (MSLQ-41)
 58. I try to think through a topic and decide what I am supposed to learn from it rather than just reading it over when studying (MSLQ-61) 59. When studying, I often try to explain the material to a classmate or friend (MSLQ-24)
60. I prefer course material that arouses my curiosity, even if it is
61. I have an uneasy, upset feeling when I take an exam (MSLQ-19) REVERSED
62. I believe that lecturers shouldn't expect students to spend significant amounts of time studying material everyone knows won't be examined (R-SPQ-2F) REVERSED
63. I would prefer to do class work that is familiar to me, rather than work I would have to learn how to do (PALS) REVERSED
64. I will do poorly because I have trouble doing assignments well (SLSQ) REVERSED
65. I wait to the last minute to start studying for upcoming tests (SRSI- SR) REVERSED
66. My aim is to avoid my teacher thinking I am a bad student (SLSQ)
67. I double-check my work to make sure what I am doing is right (SRLQ)
68. I am sure I understood the terminology used in task instructions (RLQ)
69. When studying, I often set aside time to discuss course material with a group of students from the class (MSLQ-50)
70. I find it hard to stick to a study schedule (MSLQ-52) REVERSED
71. During class time, I often miss important points because I'm thinking of other things (MSLQ-33) REVERSED
72. I will avoid participating in class if it means that other students will think I know a lot (PALS)
73. I try to apply ideas from course readings in other class activities such as lecture and discussion (MSLQ-81)
74. I ask myself questions about how well I am doing while I am learning something new (MAI)

TABLE 9. (Continued.) SMSRLQ items.

75. What I learn is relevant to me (SALES)
76. I am very interested in the content area of my courses (MSLQ-17)
77. If I don't understand the course material, it is because I didn't try nard enough (MSLQ-25)
 Whenever I read or hear an assertion or conclusion in class, I think about possible alternatives (MSLQ-71)
79. When studying, I read my class notes and the course readings over and over again (MSLQ-46)
30. One of my goals is to master a lot of new skills this year (PALS)
31. I turn a learning task into smaller, easier steps (SLSQ)
32. I'm confident I can understand the most complex material presented by the instructors (MSLQ-15)
 I keep track of how well I understand the instructor/professor (SASR)
34. I often find myself questioning things I hear or read in the courses to decide if I find them convincing (MSLQ-38)
35. When I take a test, I think about how poorly I am doing compared with other students (MSLQ-3) REVERSED
36. I change my learning strategies when I fail to understand (MAI)
37. I finish all of my studying before doing any recreational activity (SRSI-SR)
 I share material with my classmates using Dropbox or similar tools (SRLTU)
39. It's important to me that I don't look stupid in class (PALS)
90. I attend classes regularly (MSLQ-73)
11. I ask myself questions to make sure I understand the material I nave been studying (MSLQ-55)
92. What I learn encourages me to think (SALES)
33. I make simple charts, diagrams, or tables to help me organize course material (MSLQ-49)
94. I can't concentrate on a test/task when I get nervous (SASR) REVERSED
 J5. I interact outside the classroom with classmates using apps (WhatsApp, Line) and we exchange information, solve doubts, etc. (SRLTU)
26. I think the course materials are useful for me to learn (MSLQ-23)
97. When I study the readings for courses, I outline the material to help me organize my thoughts (MSLQ-32)
38. I make sure I understand before I move on to the next part of a project (SRLQ)
99. I'm a member of groups on social networks that discuss, exchange information, etc., about content-related topics (SRLTU)
100. I think of several ways to solve a problem and choose the best one (MAI)
101. I periodically review to help me understand important relationships in the courses (MAI)
102. Even if I have trouble learning the course material, I try to do the work on my own, without help from anyone (MSLQ-40) REVERSED
103. I remind myself of the steps as I am working (SLSQ)
104. When I have the opportunity, I choose course assignments that I can learn from even if they don't guarantee a good grade (MSLQ-24)
105. I tell myself to keep trying hard when I get confused (SRSI-SR)
106. I'm confident I can do an excellent job on the assignments and ests in my courses. (MSLQ-20)
107. when I study, I pull together information from different sources, such as lectures, readings, and discussions (MSLQ-53)
108. Why aim is to fully understand what I have to do (SLSQ) 109. When I take tests, I think of the consequences of failing (MSLO-
14) REVERSED 110. I make sure that I keep up with the weekly readings and
assignments for the courses (MSLQ-70) 111. I work hard to do well in class even if I don't like what we are
doing (MSLQ-48)

TABLE 9. (Continued.) SMSRLQ items.

112. I finish my work and assignments on time (SALES)
113. I share pictures on social media about course work (SRLTU)
114. I make lists of important items for the courses and memorize the lists (MSLQ-72)
115. What I learn can be used in my daily life (SALES)
116. I try to play around with ideas of my own related to what I am learning in the courses (MSLQ-66)
117. If I can, I want to get better grades than most of the other students (MSLQ-13)
118. I often find that I have been reading for the course but don't know what it was all about (MSLQ-57) REVERSED
119. I avoid asking questions in class about things I don't understand (SRSI-SR) REVERSED
120. I often find that I don't spend very much time on the courses because of other activities (MSLQ-77) REVERSED
121. I go beyond what is required to see how much I can learn (SASR)
122. I ask myself if I learned as much as I could have once I finish a task (MAI)
123. I like the subject matter of the courses I study (MSLQ-26)
124. When I study, I go over my class notes and make an outline of important concepts (MSLQ-63)
125. I ask the instructor to clarify concepts I don't understand well (MSLQ-58)
126. The most important thing for me right now is improving my overall grade point average, so my main concern is getting a good grade (MSLQ-11)
127. I feel nervous even when I am prepared for a test/task (SASR) REVERSED
128. Studying the course material will help me achieve my goals (SASR)
129. I treat the course material as a starting point and try to develop my own ideas about it (MSLQ-51)
130. When preparing for a presentation, I record myself (SRLTU)
131. I ask myself if there was an easier way to do things after I finish a task (MAI)
132. I think about what I really need to learn before I begin a task (MAI)
133. I find myself analyzing the usefulness of my learning strategies while I study (MAI)
134. I do my best so I can feel proud of myself (SLSQ)
135. What I learn is interesting (SALES)
136. I rarely find time to review my notes or readings before an exam (MSLQ-80) REVERSED
137. My aim is to pass courses while doing as little work as possible (R-SPQ-2F)
138. I often feel so lazy or bored when I study that I quit before I finish what I planned to do (MSLQ-37) REVERSED
139. I ask myself how well I accomplish my goals once I'm finished (MAI)
140. I ask my teacher about the topics that will be on upcoming tests (SRSI-SR)
141. I re-evaluate my assumptions when I get confused (MAI)
142. I spend too much time with friends when I should be studying (SASR) REVERSED
143. When course work is difficult, I either give up or only study the easy parts (MSLQ-60) REVERSED
144. When I take a test, I think about items on other parts of the test I can't answer (MSLQ-8) REVERSED

TABLE 9. (Continued.) SMSRLQ items.

145. I try to use learning strategies that have worked in the past (MAI)
146. I believe I will receive an excellent grade in my courses (MSLQ-5)
147. I look for content-related, self-evaluation exercises online and I use them to prepare for exams (SRLTU)
148. I feel my heart beating fast when I take an exam (MSLQ-28) REVERSED
149. I think I will be able to use what I learn in one course in other courses (MSLQ-4)
150. I ask myself if I have considered all options when solving a problem (MAI)

TABLE 10. SMSRLQ items.

1. This study kept me always alert and focused	
2. Interaction was adequately maintained with the lecturer	
3. I am satisfied with the quality of interaction between all involv parties in the study	ved
4. I am satisfied with my participation in the study	
5. The study encouraged me to evaluate the way in which I learn study	and
6. My knowledge about my self-learning was improved compare similar courses I studied before	d to
7. I am satisfied with the level of effort this study required	
8. The study instructions and guidelines were clearly communica me	ited to
9. Feedback on my progress during this study was given in a time manner	ely
10. Technical problems were not frequent and did not adversely a my understanding of the study	affect
11. Do you use Social networks in your daily life? If yes, what is preferred social network to communicate with your colleagues at friends?	your nd
12. Do you prefer questionnaires via Facebook or via a traditiona and paper questionnaire?	ıl pen
13. The technology used for this study was reliable	
14. Facebook is compatible for use in a study like this	
15. The use of Facebook facilitated my participation in this study	1
16. Do you prefer to be asked in a single session or throughout the whole semester? Which approach is more motivating and why?	ne

and categories. In this way, considering all of them in a single session may be overwhelming for students. Another advantage of the spaced approach is that it raises the students' SRL awareness and keeps them alert and thinking about SRL behaviors for a long time.

The results shown in the paper demonstrate that the administration of questions on a spaced basis delivers differences in the results. This can be explained by changes in the student's competences over time, or because of changes in the honesty of the answers. In any case, these results should be considered in future studies. In addition, the provision of feedback over time is in line with improvements in the acquisition of competencies and the satisfaction of students.

However, this research is not free of limitations and the results and conclusions should be confirmed in further studies. The empirical study took place in a particular semester in a few academic institutions. Since it is hard to control the students' participation via Facebook, their participation was inconsistent over the elongated period. As participation decreased with time, we eliminated the responses of 136 participants: 61 from the experimental group and 75 from the control group. Therefore, there was an uneven distribution of students, according to gender and academic year, among the experimental and control groups. Consequently, results generalization should be with caution. The decreased participation rate over time could have significantly affected the Cronbach's alpha coefficient values of the SMSRLQ. To assess this novel approach and its reliability, we need more empirical studies with larger data samples in various higher education contexts. Another limitation is the use of quantitative data only. Quantitative SRL analysis may not produce the most accurate results. For example, a student using more SRL strategies may perform worse than another who uses fewer SRL strategies more efficiently [93]. Future work shall use both qualitative and quantitative data to enrich the results by capturing the students' different SRL dimensions. We can adopt a mixed-method approach, by using self-reported data with other measures (like Learning Analytics and Educational Data Mining) to triangulate data and obtain a more comprehensive understanding [44], [72], [94].

Finally, feedback provision was a tiring and timeconsuming task for the lecturer. Therefore, there was a short delay in providing feedback to students. The faster the feedback delivery, the more influence it will have on the students' learning [95]. Future work shall include developing a conversational agent to automatize the processes of questions delivery, gathering responses, and feedback provision.

From a practical viewpoint, the findings invite higher education teachers and institutions to reflect about the importance of spaced repetition [96]. The achievement of real learning among students does not only involve punctual understanding, but also spaced reinforcement. The results obtained in this study are a new demonstration of this idea. In addition, asking different questions in class can cause serious planning problems because of the classroom time consumed. With spaced online questionnaires, students are free to answer questions at any time, leading to a different predisposition when answering. The main issue that remains to be solved is how to involve students to continue participating in the experience over time.

APPENDIX A: SMSRLQ ITEMS

See Table 9.

APPENDIX B: SATISFACTION QUESTIONNAIRE ITEMS

See Table 10.

ACKNOWLEDGMENT

The authors would like to express their deep thanks to all the involved participants. This work has received financial support from Consellería de Educación, Ciencia, Universidades e Formación Profesional and co-funded by the EU, and from the Galician Regional Government under project GPC-ED431B 2023/37.

REFERENCES

- B. J. Zimmerman, "Becoming a self-regulated learner: An overview," *Theory Into Pract.*, vol. 41, no. 2, pp. 64–70, May 2002.
- [2] E. Panadero, "A review of self-regulated learning: Six models and four directions for research," *Frontiers Psychol.*, vol. 8, pp. 1–26, Apr. 2017.
- [3] D. Lourenco and A. I. Ferreira, "Self-regulated learning and training effectiveness," *Int. J. Training Develop.*, vol. 23, no. 2, pp. 117–134, Jun. 2019, doi: 10.1111/JJTD.12149.
- [4] L. Schnaubert and D. Bodemer, "Prompting and visualising monitoring outcomes: Guiding self-regulatory processes with confidence judgments," *Learn. Instruct.*, vol. 49, pp. 251–262, Jun. 2017.
- [5] J. M. Lodge, E. Panadero, J. Broadbent, and P. G. de Barba, "Supporting self-regulated learning with learning analytics," in *Learning Analytics in the Classroom*. Evanston, IL, USA: Routledge, 2018, pp. 45–55.
- [6] R. S. Jansen, A. van Leeuwen, J. Janssen, R. Conijn, and L. Kester, "Supporting learners' self-regulated learning in massive open online courses," *Comput. Educ.*, vol. 146, Mar. 2020, Art. no. 103771.
- [7] K.-L. Lau, "The effectiveness of self-regulated learning instruction on students' classical Chinese reading comprehension and motivation," *Reading Writing*, vol. 33, no. 8, pp. 2001–2027, Feb. 2020.
- [8] R. Nejati, "The relationship between Students' self-regulated learning and reading comprehension in Iranian online classes in the COVID era," *TEL*, vol. 16, no. 2, pp. 85–109, 2022.
- [9] Y. A. S. Anwar and M. Muti'ah, "Exploration of critical thinking and selfregulated learning in online learning during the COVID-19 pandemic," *Biochemistry Mol. Biol. Educ.*, vol. 50, no. 5, pp. 502–509, Sep. 2022, doi: 10.1002/bmb.21655.
- [10] M. AL-Smadi and C. Guetl, "Supporting self-regulated learners with formative assessments using automatically created QTI-questions," in *Proc. IEEE Global Eng. Educ. Conf. (EDUCON)*, Amman, Jordan, Apr. 2011, pp. 288–294, doi: 10.1109/EDUCON.2011.5773150.
- [11] S. Hutt, J. Ocumpaugh, J. Ma, A. L. Andres, N. Bosch, L. Paquette, G. Biswas, and R. S. Baker, "Investigating SMART models of self-regulation and their impact on learning," in *Proc. EDM*, 2021, pp. 580–587.
- [12] R. S. Jansen, A. Van Leeuwen, J. Janssen, and L. Kester, "Exploring the link between self-regulated learning and learner behaviour in a massive open online course," *J. Comput. Assist. Learn.*, vol. 38, no. 4, pp. 993–1004, Aug. 2022.
- [13] T. H. Wang and C. H. Kao, "Investigating factors affecting student academic achievement in mathematics and science: Cognitive style, selfregulated learning and working memory," *Instr. Sci.*, vol. 50, no. 5, pp. 789–806, Aug. 2022.
- [14] M. Liz-Dominguez, M. Llamas-Nistal, M. Caeiro-Rodriguez, and F. Mikic-Fonte, "Learning analytics to improve the effectiveness of continuous assessment," in *Proc. Learn. Anal. Summer Inst.*, 2018, pp. 1–20.
- [15] Y. Hanafi, N. Murtadho, A. R. Hassan, M. Saefi, M. A. Ikhsan, and T. N. Diyana, "Self-regulation in qur'an learning," *Malaysian J. Learn. Instruct.*, vol. 18, pp. 103–128, Jul. 2021.
- [16] M.-T. Tran and S. Hasegawa, "Self-regulated learning recognition and improvement framework," in ACE Off. Conf. Proc., Jan. 2021, pp. 449–465.
- [17] L. Anthonysamy, "Self-regulated learning strategies for smart learning: A case of a Malaysian University," *AJRESS*, vol. 3, no. 1, pp. 72–83, 2021, doi: 10639-020-10134-2.
- [18] E. Pogorskiy, "Assessment, development and experimental evaluation of self-regulatory support in online learning," Ph.D. dissertation, School Educ., Fac. Social Sci. Health, UDUR, Durham, U.K., 2020. [Online]. Available: http://etheses.dur.ac.uk/13764/1/Pogorskiy000582351.pdf? DDD29+z
- [19] A. Bernardo, C. Galve-González, J. Nuñez, and L. Almeida, "A path model of university dropout predictors: The role of satisfaction, the use of selfregulation learning strategies and Students' engagement," *Sustainability*, vol. 14, no. 3, p. 1057, Jan. 2022.
- [20] G. Psathas, T. K. Chatzidaki, and S. N. Demetriadis, "Predictive modeling of Student dropout in MOOCs and self-regulated learning," *Computers*, vol. 12, no. 10, p. 194, Sep. 2023.

- [21] A. Tayebi, J. Gómez, and C. Delgado, "Analysis on the lack of motivation and dropout in engineering students in Spain," *IEEE Access*, vol. 9, pp. 66253–66265, 2021, doi: 10.1109/ACCESS.2021.3076751.
- [22] M. Bauer, C. Bräuer, J. Schuldt, and H. Krömker, "Adaptive E-Learning technologies for sustained learning motivation in engineering scienceacquisition of motivation through self-reports and wearable technology," in *Proc. 10th Int. Conf. Comput. Supported Educ.*, 2018, pp. 418–425.
- [23] T.-Y. Mou, "Online learning in the time of the COVID-19 crisis: Implications for the self-regulated learning of university design students," *Act. Learn. Higher Educ.*, vol. 24, no. 2, pp. 185–205, Oct. 2021, doi: 10.1177/14697874211051226.
- [24] T. M. Tran and S. Hasegawa, "An empirical study on the relationship between cognition and metacognition in technology-enhanced selfregulated learning," *Sustainability*, vol. 14, no. 7, p. 3837, Mar. 2022.
- [25] P. R. Pintrich, "A conceptual framework for assessing motivation and selfregulated learning in college students," *Educ. Psychol. Rev.*, vol. 16, no. 4, pp. 385–407, Dec. 2004.
- [26] L. Zheng, M. M. El-Bishouty, C. Pinnell, J. Bell, and V. Kumar, "A framework to automatically analyze regulation," in *Emerging Issues in Smart Learning*. Cham, Switzerland: Springer, 2015, pp. 23–30.
- [27] U. Perera Muthupoltotage and L. Gardner, "Analysing the relationships between digital literacy and self-regulated learning of Undergraduates—A preliminary investigation," in *Advances in Information Systems Development.* Cham, Switzerland: Springer, 2018, pp. 1–16.
- [28] C. R. Jackson, "Validating and adapting the motivated strategies for learning questionnaire (MSLQ) for STEM courses at an HBCU," AERA Open, vol. 4, no. 4, Oct. 2018, Art. no. 233285841880934.
- [29] R. Chakraborty, M. Haqyar, and V. K. Chechi, "Validation of motivated strategies for learning questionnaire among high school students in Afghanistan," *Pakistan J. Psychol. Res.*, vol. 36, no. 4, pp. 615–629, Jan. 2022.
- [30] A. Roth, S. Ogrin, and B. Schmitz, "Assessing self-regulated learning in higher education: A systematic literature review of self-report instruments," *Educ. Assessment, Eval. Accountability*, vol. 28, no. 3, pp. 225–250, Aug. 2016.
- [31] B. C. Choi and A. W. Pak. (2005). A Catalog of Biases in Questionnaires. [Online]. Available: https://pubmed.ncbi.nlm.nih.gov/15670466/
- [32] E. ARAKA, E. MAINA, R. GITONGA, and R. OBOKO, "A conceptual model for measuring and supporting self-regulated learning using educational data mining on learning management systems," in *Proc. IST-Africa Week Conf. (IST-Africa)*, Nairobi, Kenya, May 2019, pp. 1–11, doi: 10.23919/ISTAFRICA.2019.8764852.
- [33] M. Pammer, J. Pattermann, and S. Schlogl, "Self-regulated learning strategies and digital interruptions in webinars," in *Proc. LTEC*, 2021, pp. 51–62.
- [34] A. A. ElSayed, M. Caeiro-Rodríguez, F. A. Mikic-Fonte, and M. Llamas-Nistal, "A novel method to measure self-regulated learning based on social media," *IEEE Access*, vol. 9, pp. 93516–93528, 2021, doi: 10.1109/ACCESS.2021.3092943.
- [35] R. I. Medina-Ramírez, D. D. Álamo-Arce, F. Rodriguez-Castro, D. Cecilio-Fernandes, J. Sandars, and M. J. Costa, "Self-regulated learning microanalysis for the study of the performance of clinical examinations by physiotherapy students," *BMC Med. Educ.*, vol. 20, no. 1, pp. 1–8, Jul. 2020, doi: 10.1186/s12909-020-02149-7.
- [36] C. N. Hirt, Y. Karlen, K. M. Merki, and F. Suter, "What makes high achievers different from low achievers? Self-regulated learners in the context of a high-stakes academic long-term task," *Learn. Individual Differences*, vol. 92, Dec. 2021, Art. no. 102085.
- [37] A. F. Hadwin, P. Sukhawathanakul, R. Rostampour, and L. M. Bahena-Olivares, "Do self-regulated learning practices and intervention mitigate the impact of academic challenges and COVID-19 distress on academic performance during online learning?" *Frontiers Psychol.*, vol. 13, pp. 1–29, Mar. 2022.
- [38] A. University and B. T. Biber, "Self-regulated learning strategies used by students to prepare mathematics exams," *Int. J. Progressive Educ.*, vol. 18, no. 4, pp. 223–238, Aug. 2022.
- [39] I. Al-Dawood, "Correlation of self-regulated learning on blackboard and academic achievement of Islamic studies students," *Int. J. Learn., Teaching Educ. Res.*, vol. 21, no. 9, pp. 370–388, Sep. 2022.
- [40] R. S. Jansen, A. van Leeuwen, J. Janssen, S. Jak, and L. Kester, "Self-regulated learning partially mediates the effect of self-regulated learning interventions on achievement in higher education: A meta-analysis," *Educ. Res. Rev.*, vol. 28, Nov. 2019, Art. no. 100292.

- [41] S. Forrest, "Self-regulated learning in the IB diploma programme: A qualitative comparison of strategy use by IB diploma students with learning difficulties and those with a history of high achievement," *J. Res. Int. Educ.*, vol. 21, no. 2, pp. 139–166, Sep. 2022, doi: 10.1177/14752409221122009.
- [42] E. Panadero, J. Klug, and S. Järvelä, "Third wave of measurement in the self-regulated learning field: When measurement and intervention come hand in hand," *Scandin. J. Educ. Res.*, vol. 60, no. 6, pp. 723–735, Jul. 2015, doi: 10.1080/00313831.2015.1066436.
- [43] J.-L. Berger and S. A. Karabenick, "Construct validity of self-reported metacognitive learning strategies," *Educ. Assessment*, vol. 21, no. 1, pp. 19–33, Feb. 2016.
- [44] A. A. ElSayed, M. Caeiro-Rodriguez, F. A. MikicFonte, and M. Llamas-Nistal, "Research in learning analytics and educational data mining to measure self-regulated learning: A systematic review," in *Proc. MLearn.*, 2019, pp. 46–53.
- [45] D. Di Mitri, M. Scheffel, H. Drachsler, D. Barner, S. Ternier, and M. Specht, "Learning pulse: Using wearable biosensors and learning analytics to investigate and predict learning success in self-regulated learning," in *Proc. CrossLAK*, 2016, pp. 34–39.
- [46] J. Jovanović, D. Gasević, S. Dawson, A. Pardo, and N. Mirriahi, "Learning analytics to unveil learning strategies in a flipped classroom," *Internet Higher Educ.*, vol. 33, pp. 74–85, Apr. 2017.
- [47] A. D. Ali and W. K. Hanna, "Predicting students' achievement in a hybrid environment through self-regulated learning, log data, and course engagement: A data mining approach," *J. Educ. Comput. Res.*, vol. 60, no. 4, pp. 960–985, Jul. 2022.
- [48] S. Li, G. Chen, W. Xing, J. Zheng, and C. Xie, "Longitudinal clustering of students' self-regulated learning behaviors in engineering design," *Comput. Educ.*, vol. 153, Aug. 2020, Art. no. 103899.
- [49] J. Zheng, W. Xing, G. Zhu, G. Chen, H. Zhao, and C. Xie, "Profiling selfregulation behaviors in STEM learning of engineering design," *Comput. Educ.*, vol. 143, Jan. 2020, Art. no. 103669.
- [50] D. Pedrosa, M. M. Fontes, T. Araujo, C. Morais, T. Bettencourt, P. D. Pestana, L. Morgado, and J. Cravino, "Metacognitive challenges to support self-reflection of students in online software engineering education," in *Proc. 4th Int. Conf. Portuguese Soc. Eng. Educ. (CISPEE)*, Jun. 2021, pp. 1–10, doi: 10.1109/CISPEE47794.2021.9507230.
- [51] N. A. Uzir, D. Gasevic, W. Matcha, A. Pardo, L. A. Lim, and S. Gentili, "Discovering time management strategies in learning processes using process mining techniques," in *Proc. Eur. Conf. Technol. Enhanced Learn.*, 2019, pp. 555–569.
- [52] A. Zusho, "Toward an integrated model of Student learning in the college classroom," *Educ. Psychol. Rev.*, vol. 29, no. 2, pp. 301–324, Apr. 2017.
- [53] S. F. E. Rovers, G. Clarebout, H. H. C. M. Savelberg, A. B. H. de Bruin, and J. J. G. van Merrienboer, "Granularity matters: Comparing different ways of measuring self-regulated learning," *Metacognition Learn.*, vol. 14, no. 1, pp. 1–19, Feb. 2019.
- [54] L. McCardle and A. F. Hadwin, "Using multiple, contextualized data sources to measure learners' perceptions of their self-regulated learning," *Metacognition Learn.*, vol. 10, no. 1, pp. 43–75, Jan. 2015.
- [55] M. Liz-Domanguez, M. Caeiro-Rodroguez, M. Llamas-Nistal, and F. Mikic-Fonte, "Monitoring students' self-regulation as a basis for an early warning system," in *Proc. LASI*, 2021, pp. 38–51.
- [56] E. Araka, E. Maina, R. Gitonga, and R. Oboko, "Research trends in measurement and intervention tools for self-regulated learning for e-learning environments—Systematic review (2008–2018)," *Res. Pract. Technol. Enhanced Learn.*, vol. 15, no. 1, pp. 1–21, Mar. 2020.
- [57] A.-M. Cazan, "An intervention study for the development of selfregulated learning skills," *Current Psychol.*, vol. 41, no. 9, pp. 6406–6423, Sep. 2022.
- [58] C. Flynn, J. Olson, and M. Reinhardt, "Self-regulated learning in online graduate Bus. Communication courses: A qualitative inquiry," *Bus. Prof. Commun. Quart.*, vol. 83, no. 1, pp. 80–95, Mar. 2020, doi: 10.1177/2329490619885904.
- [59] S. Kistner, K. Rakoczy, B. Otto, C. Dignath-Van Ewijk, G. Büttner, and E. Klieme, "Promotion of self-regulated learning in classrooms: Investigating frequency, quality, and consequences for student performance," *Metacognition Learn.*, vol. 5, no. 2, pp. 157–171, Jun. 2010.
- [60] M. Mapuya, "Promoting self-regulated learning among first-year accounting-student teachers: A student-empowerment pedagogical framework," *Int. J. Learn., Teach. Educ. Res.*, vol. 21, no. 5, pp. 64–83, May 2022.

- [61] D. C. van Alten, C. Phielix, J. Janssen, and L. Kester, "Self-regulated learning support in flipped learning videos enhances learning outcomes," *Comput. Educ.*, vol. 158, Dec. 2020, Art. no. 104000.
- [62] M. Fetter, R. Robbs, and A. T. Cianciolo, "Clerkship curriculum design and USMLE step 2 performance: Exploring the impact of self-regulated exam preparation," *Med. Sci. Educator*, vol. 29, no. 1, pp. 265–276, Mar. 2019.
- [63] H. Tseng, X. Yi, and H.-T. Yeh, "Learning-related soft skills among online business students in higher education: Grade level and managerial role differences in self-regulation, motivation, and social skill," *Comput. Hum. Behav.*, vol. 95, pp. 179–186, Jun. 2019.
- [64] G. Schraw and R. S. Dennison, "Assessing metacognitive awareness," *Contemp. Educ. Psychol.*, vol. 19, no. 4, pp. 460–475, Oct. 1994.
- [65] C. Midgley, M. L. Maehr, L. Z. Hruda, E. Anderman, L. Anderman, K. E. Freeman, and T. Urdan. (2000). *Manual for the Patterns of Adaptive Learning Scales*. [Online]. Available: http://www.umich.edu/
- [66] S. Manganelli, F. Alivernini, L. Mallia, and V. Biasi, "The development and psychometric properties of the «Self-regulated knowledge scaleuniversity» (SRKS-U)," *Educ., Cultural Psychol. Stud.*, vol. 1, no. 12, pp. 235–254, Dec. 2015.
- [67] N. C. Didonato, "Effective self- and co-regulation in collaborative learning groups: An analysis of how students regulate problem solving of authentic interdisciplinary tasks," *Instructional Sci.*, vol. 41, no. 1, pp. 25–47, Feb. 2012.
- [68] A. R. Lubin, "Validation of the self-regulation strategy inventory-parent rating scale," Ph.D. dissertation, Graduate School Appl. Prof. Psychol., RU, Piscataway, NJ, USA, 2015, doi: 10.7282/T35D8TS2.
- [69] P. C. Abrami, V. Venkatesh, W. Varela, and L. Lysenko. (2012). The Student Learning Strategies Questionnaire, Version 3 (Forms A & B). [Online]. Available: https://www.concordia.ca/ content/dam/artsci/research/cslp/docs/SLSQ_FormB-post.pdf
- [70] S. Velayutham, J. Aldridge, and B. Fraser, "Development and validation of an instrument to measure students' motivation and self-regulation in science learning," *Int. J. Sci. Educ.*, vol. 33, no. 15, pp. 2159–2179, Mar. 2011.
- [71] R. F. Dugan, "Examining the construct validity of academic self-regulation using the survey of academic self-regulation (SASR)," Ph.D. dissertation, Dept. of Educational and Counseling Psychology, State Univ. New York, Albany, NY, USA, 2007.
- [72] J. Biggs, D. Kember, and D. Y. P. Leung, "The revised two-factor study process questionnaire: R-SPQ-2F," *Brit. J. Educ. Psychol.*, vol. 71, no. 1, pp. 133–149, Mar. 2001.
- [73] P. Smolen, Y. Zhang, and J. H. Byrne, "The right time to learn: Mechanisms and optimization of spaced learning," *Nat. Rev. Neurosci.*, vol. 17, no. 2, pp. 77–88, Feb. 2016.
- [74] P. R. Pintrich, D. A. Smith, T. García, and W. J. McKeachie, "A manual for the use of the motivational strategies for learning questionnaire (MSLQ)," Nat. Center Res. Improve Postsecondary Teach. Learn., Univ. Michigan, Ann Arbor, MI, USA, 1991, vol. 313, pp. 936–2741. [Online]. Available: https://eric.ed.gov/?id=ED338122
- [75] B. Tabuenca, M. Kalz, H. Drachsler, and M. Specht, "Time will tell: The role of mobile learning analytics in self-regulated learning," *Comput. Educ.*, vol. 89, pp. 53–74, Nov. 2015.
- [76] J. F. Hair, W. C. Black, B. J. Babin, and R. Anderson, *Multivariate Data Analysis*, 7th ed., New York, NY, USA: Pearson, 2010.
- [77] M. McMahon and J. Luca. (2001). Assessing Students' Self-Regulatory Skills. [Online]. Available: https://www.ascilite. org/conferences/melbourne01/pdf/papers/mcmahonm.pdf
- [78] L. Stehle and S. Wold, "Analysis of variance (ANOVA)," Chemometric Intell. Lab. Syst., vol. 6, no. 4, pp. 259–272, Nov. 1989.
- [79] R. A. Sanchez, "Student's T. Uses and abuses," *Rev. Mex. Cardiol.*, vol. 26, no. 1, pp. 59–61, 2015.
- [80] A. A. Agbo, "Cronbach's alpha: Review of limitations and associated recommendations," *J. Psychol. Afr.*, vol. 20, no. 2, pp. 233–239, Jan. 2010, doi: 10.1080/14330237.2010.10820371.
- [81] D. L. Streiner, "Starting at the beginning: An introduction to coefficient alpha and internal consistency," *J. Personality Assessment*, vol. 80, no. 1, pp. 99–103, Feb. 2003.
- [82] K. S. Taber, "The use of Cronbach's alpha when developing and reporting research instruments in science education," *Res. Sci. Educ.*, vol. 48, no. 6, pp. 1273–1296, Dec. 2018.

- [83] J. R. Segedy, J. S. Kinnebrew, and G. Biswas, "Using coherence analysis to characterize self-regulated learning behaviours in open-ended learning environments," *J. Learn. Analytics*, vol. 2, no. 1, pp. 13–48, May 2015.
- [84] H.-H. Bock, "Clustering methods: A history of k-Means algorithms," in Studies in Classification, Data Analysis, and Knowledge Organization. Cham, Switzerland: Springer, 2007, pp. 161–172.
- [85] M. Abou Naaj, M. Nachouki, and A. Ankit, "Evaluating Student satisfaction with blended learning in a gender-segregated environment," *J. Inf. Technol. Educ. Res.*, vol. 11, pp. 185–200, May 2012.
- [86] A. Bruton, J. H. Conway, and S. T. Holgate, "Reliability: What is it, and how is it measured?" *Physiotherapy*, vol. 86, no. 2, pp. 94–99, Feb. 2000.
- [87] Á. Stefkovics and Z. Kmetty, "A comparison of question order effects on item-by-item and grid formats: Visual layout matters," *Meas. Instrum. Social Sci.*, vol. 4, no. 1, pp. 1–12, Jun. 2022.
- [88] H. N. H. Cheng and X. Zhang, "Modeling the self-regulated learning behaviors of graduate students in online academic reading and writing environments," in *Advances in Analytics for Learning and Teaching*. Cham, Switzerland: Springer, 2020, pp. 227–244.
- [89] N. Sherafati and F. Mahmoudi Largani, "The potentiality of computerbased feedback in fostering EFL learners' writing performance, selfregulation ability, and self-efficacy beliefs," *J. Comput. Educ.*, vol. 10, pp. 27–55, 2023.
- [90] M. Theobald and H. Bellhäuser, "How am i going and where to next? Elaborated online feedback improves university students' self-regulated learning and performance," *Internet Higher Edu.*, vol. 55, Oct. 2022, Art. no. 100872.
- [91] L.-A. Lim, S. Dawson, D. Gasevic, S. Joksimovic, A. Fudge, A. Pardo, and S. Gentili, "Students' sense-making of personalised feedback based on learning analytics," *Australas. J. Educ. Technol.*, vol. 36, no. 6, pp. 15–33, Dec. 2020.
- [92] A. S. Labuhn, B. J. Zimmerman, and M. Hasselhorn, "Enhancing students' self-regulation and mathematics performance: The influence of feedback and self-evaluative standards," *Metacognition Learn.*, vol. 5, no. 2, pp. 173–194, Aug. 2010.
- [93] S. Vanslambrouck, C. Zhu, B. Pynoo, K. Lombaerts, J. Tondeur, and R. Scherer, "A latent profile analysis of adult students' online selfregulation in blended learning environments," *Comput. Hum. Behav.*, vol. 99, pp. 126–136, Oct. 2019.
- [94] R. Gandomkar, K. Yazdani, L. Fata, R. Mehrdad, A. Mirzazadeh, M. Jalili, and J. Sandars, "Using multiple self-regulated learning measures to understand medical students' biomedical science learning," *Med. Educ.*, vol. 54, no. 8, pp. 727–737, Aug. 2020.
- [95] A. Irons and S. Elkington, Enhancing Learning Through Formative Assessment and Feedback, 2nd ed., Evanston, IL, USA: Routledge, 2021, doi: 10.4324/9781138610514.
- [96] B. Tabibian, U. Upadhyay, A. De, A. Zarezade, B. Schölkopf, and M. Gomez-Rodriguez, "Enhancing human learning via spaced repetition optimization," *Proc. Nat. Acad. Sci. USA*, vol. 116, no. 10, pp. 3988–3993, Mar. 2019.



AMR ADEL ELSAYED was born in Cairo, Egypt, in 1982. He received the B.S. degree in business administration, major: e-commerce from the College of Management & Technology, Arab Academy for Science, Technology & Maritime Transport, Cairo, in 2009, the M.S. degree in information systems from the College of Computing & Information Technology, Arab Academy for Science, Technology & Maritime Transport, in 2014, and the Ph.D. degree in information and commu-

nication technologies from the University of Vigo, Spain, in 2023.

He is currently a full-time Lecturer with the Business Information Systems (BIS) Department, College of Management & Technology, Arab Academy for Science, Technology & Maritime Transport. His research interests include self-regulated learning (SRL), social media application in higher education, learning analytics, and big data in higher education.

MANUEL CAEIRO-RODRÍGUEZ (Senior Member, IEEE) received the Ph.D. degree in information and communication technologies in educational modeling languages, in 2007. He is currently an Associate Professor with the Department of Telematics Engineering, University of Vigo. He teaches computer programming, software engineering, and computer architecture with the School of Telecommunications Engineering. His research interests include e-learning technolo-

gies and standards, open educational resources, educational innovation, CSCL, and process-based systems. He had received several awards from the W3C, NAE CASEE new faculty fellows, and the IEEE Spanish Chapter of the Education Society to the best Ph.D. degree, in 2007. He has performed research stays with the University of Coimbra, Portugal, IRISA, Rennes, France, MTA-ZSTAKI, Budapest, Hungary, University of Kumamoto, Japan, and ISEP, Porto, Portugal. Currently, he is focused on learning analytics. He has served as the President for Spanish Chapter of the IEEE Education Society (2018–2019).



FERNANDO A. MIKIC-FONTE (Senior Member, IEEE) received the M.Sc. and Ph.D. degrees in telecommunication engineering from the University of Vigo, in 2001 and 2010, respectively. He is currently an Associate Professor with the Engineering School of Telecommunication, University of Vigo, where he is also with the Group of Telematics Systems Engineering (GIST), Telematics Engineering Department. His current research assignments, carried out with the

atlanTTic Research Center for Telecommunication Technologies, include e-learning, blended learning, self-regulated learning, serious games, and educational simulators.



ADRIÁN LUGILDE-LÓPEZ (Member, IEEE) received the M.D. degree from the Department of Telematics Engineering, University of Vigo, Spain, in 2022. He is currently pursuing the Ph.D. degree. He is also a Researcher with the atlanTTic Research Center, GIST Research Group, University of Vigo. His research interests include the application of emerging IT paradigms (virtual reality, artificial intelligence, and gamification) to the fields of e-learning and e-health.



MARTÍN LLAMAS-NISTAL (Senior Member, IEEE) received the Eng. and Ph.D. degrees in telecommunication from the Polytechnic University of Madrid, Spain, in 1986 and 1994, respectively. He has been a Faculty Member with the Higher Technical School of Telecommunication Engineers, University of Vigo, Spain, since March 1987. He is the author or co-author of more than 300 papers in peer-reviewed international refereed journals and conference proceedings. He has

directed several national and international research projects in telematics and technology enhanced learning fields. He has received several awards from the W3C and from the IEEE. He is very involved in the activities of the IEEE Education Society, serving in different positions: member of the Board of Governors (since 2008) and of the Strategic Planning Committee (since 2009), the Chair of the Publications Committee (2010–2018), the Vice-President of Publications (2011–2018), and the Vice-President of Member and Geographic Activity (2019–2020), President Elect (2021–2022), and currently he is the President (2023–2024). Since April 2020, he has been the Director of atlanTTic, the Research Center for Telecommunication Technologies, Universidade de Vigo.